Author's response to reviews

Title: The cost-effectiveness of exercise referral schemes

Authors:

Nana K Anokye (nana.anokye@brunel.ac.uk)
Paul Trueman (paul.trueman@brunel.ac.uk)
Colin Black (colin.green@pms.ac.uk)
Toby G Pavey (tgpavey@pcmd.ac.uk)
Melvyn Hillsdon (m.hillsdon@exeter.ac.uk)
Rod S Taylor (rod.taylor@pms.ac.uk)

Version: 5 Date: 20 December 2011

Author's response to reviews: see over
Dear Editor

I have made the following revisions as per your request:

1. Revised co-author name accordingly (Replace Colin Black with Colin Green)
2. Converted Appendix 1 to Table (i.e. Table 3) and hence renumbered the tables accordingly
3. Provided complete references for references 10; 12; 17; 27.

Many thanks,
Nana